

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Stein, Ernesto Hugo; Fernández Martin, Andrés; Rosenow, Samuel; Zuluaga, Victor

### Working Paper Competition-adjusted measures of real exchange rates

IDB Working Paper Series, No. IDB-WP-890

**Provided in Cooperation with:** Inter-American Development Bank (IDB), Washington, DC

*Suggested Citation:* Stein, Ernesto Hugo; Fernández Martin, Andrés; Rosenow, Samuel; Zuluaga, Victor (2018) : Competition-adjusted measures of real exchange rates, IDB Working Paper Series, No. IDB-WP-890, Inter-American Development Bank (IDB), Washington, DC, https://doi.org/10.18235/0001148

This Version is available at: https://hdl.handle.net/10419/208110

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/3.0/igo/legalcode







IDB WORKING PAPER SERIES Nº IDB-WP-890

# Competition-Adjusted Measures of Real Exchange Rates

Ernesto Stein Andrés Fernández Samuel Rosenow Victor Zuluaga

Inter-American Development Bank Department of Research and Chief Economist



## Competition-Adjusted Measures of Real Exchange Rates

Ernesto Stein\* Andrés Fernández\* Samuel Rosenow\*\* Victor Zuluaga\*

\* Inter-American Development Bank

\*\* United Nations Conference on Trade and Development

Cataloging-in-Publication data provided by the Inter-American Development Bank Felipe Herrera Library

Competition-adjusted measures of real exchange rates / Ernesto Stein, Andres Fernandez, Samuel Rosenow, Victor Zuluaga.

p. cm. — (IDB Working Paper Series ; 890)

Includes bibliographic references.

1. Foreign exchange rates-Econometric models. 2. Competition, International-Econometric models. 3. International trade-Econometric models. I. Stein, Ernesto. II. Fernandez, Andres. III. Rosenow, Samuel. IV. Zuluaga, Victor. V. Inter-American Development Bank. Department of Research and Chief Economist. VI. Series. IDB-WP-890

#### http://www.iadb.org

Copyright © 2018 Inter-American Development Bank. This work is licensed under a Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives (CC-IGO BY-NC-ND 3.0 IGO) license (<u>http://creativecommons.org/licenses/by-nc-nd/3.0/igo/</u> <u>legalcode</u>) and may be reproduced with attribution to the IDB and for any non-commercial purpose, as provided below. No derivative work is allowed.

Any dispute related to the use of the works of the IDB that cannot be settled amicably shall be submitted to arbitration pursuant to the UNCITRAL rules. The use of the IDB's name for any purpose other than for attribution, and the use of IDB's logo shall be subject to a separate written license agreement between the IDB and the user and is not authorized as part of this CC-IGO license.

Following a peer review process, and with previous written consent by the Inter-American Development Bank (IDB), a revised version of this work may also be reproduced in any academic journal, including those indexed by the American Economic Association's EconLit, provided that the IDB is credited and that the author(s) receive no income from the publication. Therefore, the restriction to receive income from such publication shall only extend to the publication's author(s). With regard to such restriction, in case of any inconsistency between the Creative Commons IGO 3.0 Attribution-NonCommercial-NoDerivatives license and these statements, the latter shall prevail.

Note that link provided above includes additional terms and conditions of the license.

The opinions expressed in this publication are those of the authors and do not necessarily reflect the views of the Inter-American Development Bank, its Board of Directors, or the countries they represent.



### Abstract\*

We develop a methodology to construct real effective exchange rates that incorporate two distinctive elements not accounted for in the traditional measures: i) competition in third markets and ii) adjustments for similarity in export baskets between exporters and their competitors. In addition to constructing competitionadjusted real effective exchange rates at the aggregate country level, we develop similar measures at the country-product, country-destination, and country-productdestination level. We then build a novel and public dataset where we apply this methodology to compute monthly adjusted REERs for a panel of 120 countries and 769 products. As an application, we use the dataset to examine the changes in export competitiveness in countries in Latin America and the Caribbean between May 2014 and February 2016, a period characterized by substantial movements in exchange rates. We find that using traditional measures of real effective exchange rates misallocates between one third and one half of the relevant weights, and it leads to an important underestimation of the loss in export competitiveness. Furthermore, we find that there are very significant differences across products and destinations with regards to changes in export competitiveness.

**JEL classifications:** F10, F31

Keywords: Real effective exchange rate, Competitiveness, Trade

<sup>&</sup>lt;sup>\*</sup> This paper benefited from helpful comments from participants in seminars at the Inter-American Development Bank. Lorena Caro, Rodrigo García and Jaime Ramírez provided excellent research assistance. The views expressed by the authors in this paper do not reflect the opinions of the Inter-American Development Bank, its Board of Directors or the countries it represents. Corresponding author: Ernesto Stein (<u>ernestos@iadb.org</u>).

### 1 Introduction

The real effective exchange rate (REER) is the most commonly used measure for assessing a country's international competitiveness. It tracks the evolution of price (or cost) competitiveness of a country with respect to its trading partners. Traditionally, REERs were calculated as the geometric weighted average of bilateral real exchange rates between pairs of countries, using trade shares as weights. In this paper, we deviate from the traditional measure. First, we develop a methodology for computing REERs at the aggregate country level that incorporates two elements not accounted for in the traditional measure: i) competition in third markets and ii) adjustments for similarity in export baskets between exporters and their competitors. While other authors have developed alternative competition-adjusted REERs at the aggregate country level (see, for example, McGuirk, 1986; Zanello and Desruelle, 1997; Bayoumi et al., 2006), our measures introduce novel elements into the computation of third market competition-adjusted real effective exchange rates at the aggregate country level, we extend the methodology to develop measures at the product, destination and product-destination level. To the best of our knowledge, none of these more disaggregated competition-adjusted REERs have been developed before.

The third contribution is empirical, as we build a new dataset where we apply this methodology to compute adjusted REERs at the country, country-product and country-destination levels using data on exchange rates, prices, production and bilateral trade for 120 countries and 769 products at the 4-digit Standard International Trade Classification (SITC) level (Revision 2). We make this dataset publicly available. Lastly, to illustrate the usefulness of our methodological contribution, as well as to highlight an application of the new dataset, we show the impact of these adjustments on the exchange rate weights and patterns for countries in Latin America and the Caribbean between May 2014 and February 2016 at the aggregate, product, destination and product-destination levels. This period is convenient to explore the relevance of using our competition-adjusted real effective exchange rates, as it was characterized by widespread and large nominal as well as real exchange rate depreciations vis-à-vis the US dollar.

Our results show that the weights for the adjusted REER measure are significantly different from the traditional weights of REERs that leave out the competition in third markets, leading to important differences in the evolution of countries' export competitiveness. On average, 49 percent of the weights corresponding to Latin America and the Caribbean countries shift as a result of the adjustment, while 44 percent of the weights shift in the case of the world sample.<sup>1</sup>Moreover, different products within countries exhibit very different experiences in terms of their exchange rate competitiveness, justifying our effort to develop REERs at the product level. Similarly, we find that our destination-level competition-adjusted REERs differ substantially from the more traditional bilateral real exchange rates, which only consider the evolution of prices and nominal exchange rates in each pair of countries. Concretely, for countries in Latin America and the Caribbean, the weight assigned to each destination country is, on average, 0.45 as opposed to one as in the traditional bilateral real exchange rate case.<sup>2</sup> The same figure is 0.37 for the complete sample.

The economic intuition as to why the two adjustments that we develop help refine the assessment of a country's international competitiveness is simple. Traditional measures that use trade shares as weights

<sup>&</sup>lt;sup>1</sup>This means that nearly half of the weights used to calculate traditional REERs are reallocated to other trading partners when adjusting for third-country competition and export basket similarity.

<sup>&</sup>lt;sup>2</sup>In other words, in calculating the competition-adjusted REER of country A in country B, exporters from country A face competition from producers in country B (accounting on average for less than half of the weight) and exporters from third countries (accounting on average for more than half of the weight). In contrast, traditional measures only consider the destination country (and thus assign all the weight to this country) when computing bilateral REERs.

make an important implicit assumption: when a country exports a good to a specific trading partner, it is competing only with producers in the destination market. This assumption, however, is problematic (see, for example, McGuirk, 1986; Chinn, 2006). When Mexican producers export color TVs to the United States, they are not just competing with US producers. They are also competing with Chinese manufacturers. While China may not be an important export destination for Mexico, it is an important competitor in third markets. Thus, it makes sense for the Chinese currency to have a significant weight in the calculation of Mexico's real effective exchange rate. If two countries had completely different export baskets, however, the fact that they both export to the same third country would not imply that they actually compete. This is where the adjustment for export basket similarity comes in. The fact that Mexico and China have fairly similar export baskets makes competition in third markets between these two countries more relevant.

**Related Literature** - The idea of taking into account competition in third markets is not new. The earliest efforts in this regard go back to the Multilateral Exchange Rate Model (MERM) of the IMF. The MERM, first presented in Artus and Rhomberg (1973) and then refined in Artus and McGuirk (1981), was a macroeconomic model set up to assess the link between countries' real effective exchange rates and resulting current account balances. The MERM-weighted index departed from the traditional trade-based weights used at the time. Instead, countries were weighted in proportion to the impact that, according to the MERM model, a 1 percent increase in the price of each foreign currency would have on the home country trade balance. In addition to considering whether countries competed in third markets in very broad categories of products (such as semifinished manufactures, SITC 5-6, or finished manufactures, SITC 7-9), MERM considered price elasticities of demand and supply for different types of products, relying on the Armington (1969) assumption that similar goods from different countries are imperfect substitutes, and thus face finite elasticities of demand. MERM-weighted REERs and REERs based on similar models were influential in their time. They were reported in the IMF's International Financial Statistics (IFS) for 18 developed countries between 1973 and 1989, and Rhomberg (1976) reports that they were also the basis for REER data published by the UK Treasury, the Bank for International Settlements, the Council of Economic Advisors, and the OECD. However, as reported by Boughton (1997), the static nature of the MERM model made it obsolete in the mid-1980s. The MERM model was phased out, and with it, MERM-weighted REER indices were lost as well.

The IMF continued to be at the forefront of efforts to measure real effective exchange rates including competition in third markets. In 1983, the Fund created the Information Notice System (INS) to fulfill the mandate stipulated in Article IV to "exercise firm surveillance over the exchange rate policies of members" (Zanello and Desruelle, 1997). Real effective exchange rates calculated for the INS, based on theoretical work by McGuirk (1986), as well as changes over time in country coverage and methodology, are discussed in detail in Zanello and Desruelle (1997) as well as in Bayoumi et al. (2006). To date, the latter represents the Fund's current methodology to calculate REERs, which are published regularly in the IFS. Commodities are treated as global goods, where countries compete at the level of the world market, rather than at the country market level. For the case of manufacturing, the Fund incorporates competition in the domestic (import) market and double export weights, accounting for competition with each trading partner in that partner's market, as well as in third markets. REER methodologies used by the Bank for International Settlement (Klau, 2006) or the European Central Bank (Buldorini, 2002) are similar in flavor to that of the Fund.

While our work is motivated by these earlier contributions, it also deviates from them in important ways. The first difference relates to the way in which we introduce domestic production into the calculation of our adjusted REERs. When considering, for example, Mexican exports of manufactures to the United States, any approach that incorporates competition in third markets will need to establish the extent to which Mexican exporters of manufactures to the United States are facing competition from US domestic producers, vis-à-vis producers from other countries such as South Korea or China. In order to distribute weights appropriately, it is necessary to compare US domestic production of manufactures to imports of manufactures from third countries. The existing REERs discussed above use gross output as a measure of domestic production. In doing so, intermediate inputs used in the production of manufactures may be counted multiple times, leading to an overestimation of the weight of the destination country (in this example, the United States), and thus an underestimation of the third-market competition effect. In the specific case of Mexican exports to the United States, competition from third markets is underestimated by 15 percent. By using value added production data instead, we avoid this undesirable feature of competition-adjusted REERs.<sup>3</sup>

Second, another important shortcoming of existing REERs is that there is no attempt to differentiate across different types of manufactures, which are treated as a representative product. Thus, if Germany exports cars and Honduras exports garments to the US market, insofar as both products are classified as manufactures, producers of these dissimilar products would be considered direct competitors. The implicit assumption is that the elasticity of substitution between German cars and Honduran shirts is equal to that between German shirts and Honduran shirts. By introducing an adjustment for similarity of export basketsand using disaggregated trade data at the 4-digit SITC level, we move away from this undesirable representative good approach.<sup>4,5</sup>

Third, while existing measures are only available at the aggregate country level, we develop competitionadjusted REERs at the country-product, the country-destination, and the country-product-destination level, in addition to our aggregate country level measures. Fourth, our competition-adjusted REERs are available for a wider set of countries (120), compared to those in that are reported in the IFS database (94). In particular, in the case of Latin America and the Caribbean countries we use in our analysis, our data cover 23 countries, while those in the IFS covers 13.

The rest of the paper is structured as follows. Section 2 explains the methodology used to compute the AREER at the country and country-product level as well as the country-destination and country-product-destination level. Section 3 describes the datasets that are publicly available as well as the main sources used in building them. Section 4 presents the analysis of the 2014-16 episode of widespread nominal exchange rate depreciations in Latin America and the Caribbean. Finally, Section 5 concludes, and discusses policy implications and avenues for further research.

<sup>&</sup>lt;sup>3</sup>Recently, some authors such as Bems and Johnson (2017), Bayoumi et al. (2013) and Patel et al. (2014) have proposed formulas that focus directly on trade in value added, reflecting the increasing importance of global value chains. This is an important consideration, which accounts for the fact that imports from one country could in fact contain inputs produced in another, something that is not captured in our measures that assign weights depending on the country from which products are directly imported. The input-output data necessary to implement these formulas have limited country coverage, particularly in the Latin American and the Caribbean countries that we use in the application of our methodology. We hope to be able to address this important issue in the future, as more data become available.

 $<sup>^{4}</sup>$ By adjusting for export basket similarity, we are assuming that the elasticity of substitution between cars and shirts, or any other pair of goods for that matter, is zero, regardless of country of origin. While this may be an extreme assumption, we believe that it is more realistic than the alternative.

<sup>&</sup>lt;sup>5</sup>Bennett and Zarnic (2009)also departs from this representative product approach. Instead, it introduces what the authors call a heterogeneous product approach, which defines market competition at the 4-digit International Standard Industrial Classification (ISIC) product level. But they apply this approach to study REERs of just four southern European countries, namely Greece, Italy, Portugal and Spain. Our work, in contrast, covers a much wider range of developed and developing countries.

### 2 Adjusted Real Effective Exchange Rates

### 2.1 AREER at the Country Level

### 2.1.1 Adjusting for Competition in Third Markets

The real effective exchange rate of a country is generally defined as the geometric average of its bilateral real exchange rates with each of its trading partners.<sup>6</sup> More formally,

$$REER_i = \prod_{j=1}^n \left(\frac{\epsilon_{ij}P_i}{P_j}\right)^{w_{ij}} = \prod_{j=1}^n RER_{ij}^{w_{ij}},\tag{1}$$

where  $\epsilon_{ij}$  is the nominal exchange rate between countries *i* and *j* (price of *i*'s currency in terms of *j*'s currency), *n* is the number of *i*'s potential trading partners,  $P_i$  and  $P_j$  are the price levels in local currency of countries *i* and *j*,  $RER_{ij}$  is the real bilateral exchange rate between countries *i* and *j*, and  $w_{ij}$  is the weight that corresponds to country *j* in the calculation of country *i*'s REER. Notice that we have defined the REER so that an increase is an appreciation.<sup>7</sup> Traditional measures of REER use export shares as weights:<sup>8</sup>

$$w_{ij}^{tr} = \frac{X_{ij}}{\sum_{j} X_{ij}} = \frac{X_{ij}}{X_{i.}},$$
(2)

where  $X_{ij}$  are exports from *i* to *j*, and  $X_i$  are total exports from *i*. Thus, if 71.4 percent of all Mexican exports have the United States as destination,  $w_{MEX,US}^{tr}$  would be equal to 0.714 in the calculation of Mexico's REER. As noted in the introduction, however, this traditional measure has serious shortcomings. As the example of Mexican color TV producers makes clear, when a country exports a product to a destination, exporters are not just competing with producers of that product in that destination. They also compete with producers from other countries who export to the same destination. In order to address this problem, we propose a measure of REER that takes into account not just export shares, but also competition in third markets.<sup>9</sup> We call this measure adjusted real effective exchange rate, or AREER.

To build on the intuition on how our AREER works, it is helpful to go back to the example of the share in Mexico's AREER that corresponds to Mexican exports to the US market. Rather than assigning a weight of 0.714 to the United States, as would be the case with the traditional measure, the share corresponding to the US market is divided into two portions. One portion, which we denote  $\alpha_{US,MEX}$ , representing the share of US absorption of (non-Mexican) tradables satisfied by domestic producers, is still assigned to the United States. The rest,  $(1 - \alpha_{US,MEX})$ , corresponding to the share of imports in the US absorption of (non-Mexican) tradables, is assigned to countries other than Mexico that export to the United States, in proportion to their export shares. More formally,  $\alpha_{ji}$  is defined as

$$\alpha_{ji} = \frac{Y_j - \tau_j X_j}{Y_j - X_j + M_{j,-i}},$$
(3)

where  $Y_j$  is the value added of tradables in country j,  $\tau_j$  is the percentage of domestic value added in country

<sup>&</sup>lt;sup>6</sup>When calculating effective exchange rates it is standard practice to use geometric averages rather than arithmetic averages, which have undesirable features. For example, while a percentage change in the geometrically averaged effective exchange rate between two periods is independent of the base period, the choice of the base period affects REERs when an arithmetic average is applied.

<sup>&</sup>lt;sup>7</sup>Throughout the discussion, the time subindex will be omitted for simplification.

<sup>&</sup>lt;sup>8</sup>Some measures of REER consider the shares in both exports and imports. Given the focus on export competitiveness, in this paper we will use export shares only.

 $<sup>^{9}</sup>$ As discussed in the introduction, other authors such as McGuirk (1986) have already introduced REERs that take into account competition in third markets, based on Armington's (1969) demand system.

j's exports  $(X_j)$ , and  $M_{j,-i}$  are country j's imports that do not originate in country i  $(\sum_{k\neq i}^n M_{jk}, where M_{jk})$  are the imports from country k to j). Thus, the denominator,  $(Y_j - X_j + M_{j,-i})$ , is the total absorption of tradables in country j that does not originate in country i, and  $\alpha_{ji}$  is the proportion of this absorption that is locally sourced. Notice that, in the numerator of (3), we only subtract from the production of tradables the portion of exports of country j that corresponds to local value added. For example, if a country imports car engines and exports cars, the engines are not accounted for in  $Y_j$ , and thus should not be counted either when substracting total exports in the numerator. Figure 1 provides intuition for this calculation for the case of the United States, showing how the relevant  $\alpha$  for the calculation of Mexico's AREER, that is equal to 54 percent ( $\alpha_{US,MEX}$ , as shown in panel a of the Figure), differs from that relevant to Colombia, that is 50.6 percent ( $\alpha_{US,COL}$ , as shown in panel b), due to the Mexican share in US imports being larger than that of Colombia (13 percent for Mexico versus 1 percent for Colombia).

### [Figure 1 about here.]

Continuing with the Mexico-United States example, the weight of the United State corresponding to its role in the US market is then  $\alpha_{US,MEX} w_{MEX,US}^{tr}$ . But in order to compute the weight of the United States in Mexico's AREER, which we call  $w_{MEX,US}^1$ , we need to add a second component, which accounts for the weight of the United States in all other Mexican export markets. Thus, more generally, the total weight of country j in country i's competition adjusted REER will be given by

$$w_{ij}^{1} = \alpha_{ji} \frac{X_{ij}}{X_{i}} + \sum_{\substack{k=1\\k\neq i,j}}^{n} (1 - \alpha_{ki}) \frac{X_{ik}}{X_{i}} \frac{M_{kj}}{M_{k,-i}} = \alpha_{ji} w_{ij}^{tr} + \sum_{\substack{k=1\\k\neq i,j}}^{n} (1 - \alpha_{ki}) w_{ik}^{tr} \frac{M_{kj}}{M_{k,-i}}.$$
(4)

The first term in equation (4) captures the direct competition between producers from countries i and j in market j. The second term captures the competition between producers from country i and producers from country j in third markets k. Notice that equations (2) and (4) are equivalent when all  $\alpha_{ji}$  are equal to one. In that case, the AREER reduces to the traditional REER.

Table 1 illustrates graphically how this works for Mexico. Countries k in the top row are the most important destinations for Mexican exports. The percentages immediately below represent the importance of each market in Mexico's exports, and are the weights corresponding to the traditional REER calculation. For instance, 71.4 percent of total Mexican exports go to the United States, 6.5 percent to Canada, and only 2.7 percent to China. Countries j in the rows represent Mexico's competitors in each destination market k. The percentages in the diagonal (where j = k) represent the alphas for each of these destinations, from the point of view of Mexican exporters (that is, the  $\alpha_{i,MEX}$ ).

#### [Table 1 about here.]

Let us consider the United States' weight in Mexico's AREER depicted in the first row. The first cell, 54 percent, corresponds to  $\alpha_{USA,MEX}$  and represents the share of US absorption satisfied by its domestic producers (Figure 1 panel a). The second cell, 36.9 percent, is the share of the United States in Canada's non-Mexican absorption of tradables. This comes from multiplying  $(1 - \alpha_{CAN,MEX})$ , which is equal to 64.4 percent, by the US share in non-Mexican Canadian imports or  $\frac{M_{CAN,USA}}{M_{CAN,-MEX}}$ , which is 55.7 percent. Similarly, the third cell (2.7 percent) represents the US share in China's non-Mexican absorption, and so forth. The total weight corresponding to the United States in the calculation of the AREER of Mexico is obtained by multiplying each of the cells of the United States' row by the importance of each destination market in

Mexico's exports and adding these horizontally. Thus, the United States' weight would be 54% \* 71.4% + 36.9% \* 6.5% + 2.8% \* 2.7% + ... and so on, for a total weight of 42.3 percent, as indicated in the last column. The first term of this summation (54% \* 71.4% = 38.6%) corresponds to the first term in equation (4), that is, the weight of US producers as competitors of Mexican exporters in the US market alone. The rest (3.7 percent) corresponds to the second term in equation (4) or the role of the United States as a competitor to Mexican exporters in third markets. A comparison of the adjusted and the traditional weights for the United States (42.3 percent vs. 71.4 percent) clearly shows that the adjustment we make by taking into account third market competition is not trivial. In contrast to this fall in the weight of the United States, a country like China significantly increases its weight, going from 2.7 percent in the traditional case to 12.2 once we account for competition in third markets. Obviously, Chinese producers are much more important than US producers as competitors of Mexican producers in third markets.

As can be seen from the discussion of equation (3), our measure of  $\alpha_{ji}$ , the share of local absorption of tradables, and thus our AREERs, use production data in value-added terms. This differs from the REERs reported by the IMF, the BIS, or the ECB, which use gross output instead, under the implicit assumption of no intermediate inputs. Once intermediate inputs are taken into account, using gross output results in double counting.<sup>10</sup> By using production in value-added terms, we avoid this double counting, which can represent an important portion of gross output, and would lead to an overestimation of the  $\alpha_{ji}$ . For instance, using information from the World Input-Ouput Database (Timmer et al., 2015) for the year 2013, the share of manufacturing gross output that corresponds to intermediate inputs from the local manufacturing sector was 33.6 percent for Brazil, 42.3 percent for Mexico, and 32.7 percent for the United States. For the latter country, this would lead to an  $\alpha_{US,MEX}$  of 60.8 percent rather than 54 percent, an overestimation of 12.6 percent.

### 2.1.2 Adjusting for Similarity of Export Baskets

So far, we have addressed one important problem with traditional REER measures: they do not account for competition in third markets. However, we have assumed that two countries exporting to the same destination compete between them, regardless of the composition of their export baskets. This is clearly an undesirable assumption. Two countries may export completely different products to the same destination. In that case, they are not really competing. In this section, we introduce an additional adjustment to our REER weights in order to account for export basket similarity. We do so using an index developed by Finger and Kreinin (1979), which defines export similarity  $S_{ij}$  between country *i* and *j* across a basket of *P* goods as follows:

$$S_{ij} = \sum_{p=1}^{P} \min\left(\frac{x_{ip}}{X_i}, \frac{x_{jp}}{X_j}\right),\tag{5}$$

where  $x_{ip}/x_i$  represents the share of good p in country *i*'s total exports and P is the total number of products exported by at least one country in the sample of n countries. Obviously, the level of aggregation matters in this case. As we will discuss in more detail in section 3, we use data on exports at the four digit SITC Rev. 2 level.  $S_{ij}$  takes higher values the more similar the participation of each product p is in the export basket between *i* and *j*. To obtain the final weights for each country for the AREER, adjusting for both competition and similarity, the competition-adjusted weights are multiplied by the similarity index. Since

<sup>&</sup>lt;sup>10</sup>For example, if a firm in the United States manufactures microchips and another uses these to manufacture computers, the value of the microchips will be counted twice when measuring gross manufacturing output.

the similarity index varies between 0 and 1, as a result of the multiplication the sum of the weights for the AREER computation in country i will no longer be 1. For this reason, the weights are renormalized so that they add up to 1:

$$w_{ij}^2 = \frac{S_{ij} w_{ij}^1}{\sum_j S_{ij} w_{ij}^1}.$$
 (6)

Once we adjust for competition in third markets and similarity, the adjusted real effective exchange rate becomes

$$AREER_i = \prod_{j=1}^n RER_{ij}^{w_{ij}^2}.$$
(7)

#### 2.1.3 Manufacturing AREER

So far it has been assumed that equation (4) applies to all tradable goods, including not only manufactures but also commodities. As with the MERM model discussed above, this equation also relies on the Armington (1969) assumption that similar goods from different countries are imperfect substitutes, and thus face finite elasticities of demand. While this seems like a reasonable assumption for manufactures, which tend to be differentiated, it may not be ideal for the case of commodities. If prices of homogeneous goods are set in integrated global markets, then two countries producing the same commodity could be considered to be competing, regardless of whether they sell them in different markets. In order to address this issue, we produce two additional versions of the AREER index. The first one, AREER-M, leaves commodities out, and calculates weights focusing exclusively on trade of manufactured products (thus the M). The second one, which we will discuss in the next subsection, follows Bayoumi et al. (2006) and combines different sets of weights for manufacturing and commodities into a single composite index.

An equation similar to (4), but focused exclusively on manufactured trade flows, could in principle be used to obtain the weights corresponding to AREER-M,  $w_{ij}^m$ . However, it is necessary to modify  $\alpha_{ji}$ , the share of absorption of tradables satisfied by domestic producers, in the following way:

$$\alpha_{ji}^{m} = \frac{Y_{j}^{m} - \tau_{j}^{m} \kappa_{j} X_{j}^{m}}{Y_{j}^{m} - \left(\tau_{j}^{m} \kappa_{j} + \left(1 - \tau_{j}^{m}\right) \overline{\kappa}_{-j}\right) X_{j}^{m} + M_{j,-i}^{m}}.$$
(8)

Recall that in the numerator of equation (3) we multiplied exports by their share of domestic value added  $\tau_j$  before subtracting these exports from the total value added in tradables  $Y_j$ . In calculating the weights for manufacturing, we face the following problem: we would like to be subtracting manufactured local value added in manufacturing exports. But  $\tau_j^m$  is total local value added in manufacturing exports. If a country exports cars and part of the local value added is natural rubber for the tires, we face the problem that the rubber would be included in the second term of the numerator, but not in the first (since natural rubber production is not part of  $Y_j^m$ ). In order to exclude natural rubber and other non-manufactures from the second term, we introduce an additional factor,  $\kappa_j$ , which measures the participation of the manufacturing sector in total value added of manufacturing in the export value of that same sector in country j. Thus,  $\tau_j^m \kappa_j X_j^m$  approximates the value added by the local manufacturing sector in the manufacturing exports of j.

An additional adjustment needs to be done in the denominator in order to obtain the total absorption of manufactures. Consider a country that imports steel to produce and export cars. Imports of steel are not accounted for in  $M_{j,-i}^m$  but, without the necessary adjustment, would be counted as part of the exports of cars. To solve this problem, we multiply  $(1 - \tau_j^m)$ , the non-local portion of value added in exports, by  $\overline{\kappa}_{-j}$ , which captures the average share of the manufacturing sector in value added in countries where manufacturing imports from j originate, weighted by manufacturing import shares. In this case,  $(1 - \tau_j^m) \overline{\kappa}_{-j} X_j^m$  measures the proportion of manufacturing exports of j added through manufacturing imports from other countries.<sup>11</sup>

#### 2.1.4 A Composite AREER

In this subsection we present a variation of the index that combines different weights for trade in manufactures and commodities into a single composite index, which we denote AREER-MC. Following Bayoumi et al. (2006) the total weight corresponding to country j for country i's exchange rate is:

$$w_{ij}^{mc} = \lambda_i^m w_{ij}^m + \lambda_i^c w_{ij}^c, \tag{9}$$

where  $\lambda_i^m$  and  $\lambda_i^c$  represent the share of manufacturing and commodities in the export basket of  $i^{12} w_{ij}^m$  is calculated as discussed in the previous section, and  $w_{ij}^c$  is defined as in Bayoumi et al. (2006):

$$w_{ij}^{c} = \sum_{c} \frac{X_{j}^{c}}{\sum_{\substack{k=1\\k\neq i}}^{n} X_{k}^{c}} \frac{X_{i}^{c}}{\sum_{c} X_{i}^{c}}.$$
(10)

The weight of country j for the calculation of commodity AREER in country i is the sum over all the commodities of two factors. The first one is the share of country j in the total world exports of that commodity (excluding those of country i itself),  $X_{j}^{c}/\sum_{k=1}^{n} X_{k}^{c}$ . The second one is the share of that commodity in total commodity exports of country i,  $X_{i}^{c}/\sum_{c} X_{i}^{c}$ . Note that equation (10) assumes that competition in commodity markets occurs globally.<sup>13</sup>

### 2.2 AREER at the Country-Product Level

The methodology used to calculate AREER at the country and sector levels can also be used to calculate country-product-specific AREERs. Producers of different products within a country export to different destinations, where they compete with exporters of different origins. Thus, the evolution of export competitiveness in a country can vary significantly across products, even within the same sector. Consider for example the case of coffee and flower exporters in Colombia, a country whose currency depreciated substantially since mid-2014 until early 2016. In both cases, the most important destination market is the United States. In coffee, the main competitor is Brazil, another country that depreciated substantially. In contrast, the main competitor in flowers is Ecuador, a dollarized economy which experienced considerable real appreciation as the US dollar strengthened within that period. Thus, producers of these two products experienced substantially different changes in exchange rate competitiveness in the last few years. Calculating AREERs at the country-product level will allow us to document these issues, as well as study the impact of real exchange rates on exports at the product level in future work. We are not aware of any papers that have

<sup>&</sup>lt;sup>11</sup>Notice that an alternative to the equation (8) would have been to define  $\tau_j^m$  as the share of the value added in manufacturing exports originated in the domestic manufacturing sector. We did not adopt this strategy because it requires detailed "global" input-output tables that are generally available for a limited group of countries, most of them developed ones (OECD, 2013; Timmer et al., 2015).

<sup>&</sup>lt;sup>12</sup>Following Bayoumi et al. (2006), we defined manufacturing and commodities such that  $\lambda_i^m$  and  $\lambda_i^c$  add up to one.

<sup>&</sup>lt;sup>13</sup>Unlike the equation used by Bayoumi et al. (2006), which does not exclude exports from country i in the denominator of the first factor, the expression (10) adds up to one for all i. Thus, it is not necessary to perform any additional normalization to the weights, as done by these authors.

proposed real exchange rate measures at the product level, whether using traditional or adjusted weights. In that regard, we see this as an important contribution of this paper.

Taking equation (1) as a starting point, we can calculate the REER at the country-product level. In particular,

$$REER_{ip} = \prod_{j=1}^{n} RER_{ij}^{w_{pij}},\tag{11}$$

where  $w_{pij}$  represents the weight that country j receives in the calculation of the REER of country i for product p. Analogue to the REER at the country level,  $w_{pij}$  can be calculated in the traditional way, that is,

$$w_{pij}^{0} = \frac{x_{pij}}{\sum_{j} x_{pij}} = \frac{x_{pij}}{x_{pi}}.$$
 (12)

In this case, the weights correspond to the share of exports to each destination country in country i's total exports of good p.<sup>14</sup> As in the calculation of traditional REERs at the country level, these weights implicitly assume that producers from country i only compete with producers in the destination country. Departing from this undesirable assumption, it is possible to calculate product-specific weights that adjust for competition in third markets:

$$w_{pij}^{1} = \alpha_{pji} \frac{x_{pij}}{x_{pi}} + \sum_{\substack{k=1\\k \neq i,j}}^{n} (1 - \alpha_{pki}) \frac{m_{pkj}}{m_{pk,-i}} \frac{x_{pik}}{x_{pi}},$$
(13)

with  $\alpha_{pji}$  representing the share of j's domestic production in its total absorption of good p (not originating in i). Note that the equation does not include any adjustment for export similarity since it is now defined at a high level of product disaggregation.

Following the discussion on manufactures, it is possible to define  $\alpha_{pji}$  as follows:

$$\alpha_{pji} = \frac{y_{pj} - \tau_{pj}\kappa_{pj}x_{pj}}{y_{pj} - \tau_{pj}\kappa_{pj}x_{pj} + m_{pj,-i}}$$
(14)

where  $y_{pj}$  represents the value added of product p in j,  $\tau_{pj}$  the share of exports of that product which is added locally, and  $\kappa_{pj}$  the participation of industry p in the local value added of j's exports of that product. Notice that in both the numerator and denominator it is necessary to include  $\kappa_{pj}$  because  $\tau_{pj}x_{pj}$  captures the contribution of all domestic sectors in j's exports of product p, whereas  $y_{pj}$  only captures the value added of p in j (without counting the contribution of other sectors within country j).<sup>15</sup> Notice, that contrary to (8), this equation does not correct by the value added in exports of p that is imported from within the sector (i.e., the factor  $(1 - \tau_j^m) \overline{\kappa}_{-j} X_j^m$  in equation (8)), under the reasonable assumption that, at the product level, p is not used as input in its own production (i.e., you do not use imported cars as an input to produce cars).

Computing  $\kappa_{pj}$  requires detailed input-output data comparable across countries and products that is only

 $<sup>^{14}\</sup>mathrm{We}$  use lowercase letters for variables disaggregated to the product level.

<sup>&</sup>lt;sup>15</sup>Going back to the example of the auto industry,  $x_{pj}$  would be country j's exports of cars.  $\tau_{pj}$  captures the domestic content of the exports of cars (for example, if the engines are imported, it would capture the value of the car minus the value of the engine in proportion to the value of the car). But if the production of cars in country j also involves buying domestic leather produced in a different industry, then  $\kappa_{pj}$  captures the proportion of the domestic content of car exports that is added exclusively within the car industry. In other words, in this example  $\kappa_{pj}$  would be the value of the car minus the value of the (domestic) leather, all divided by the value of the car plus the value of the leather.

available for a small subset of countries. However, under the (admittedly strong) assumption that the ratio between the gross output and value added of good p is equal to  $\xi_p$  for all origins, it can be shown that  $\alpha_{pji}$ can be rewritten as follows (see mathematical appendix):

$$\alpha_{pji} = \frac{y_{pj}^* - x_{pj}}{y_{pj}^* - x_{pj} + \xi_p m_{pj,-i}},\tag{15}$$

with  $y_{pj}^*$  representing the gross output of good p in country j, and  $m_{pj,-i}$  the imports of j (not originating in i) of the same good. In contrast to equation (14), this expression for  $\alpha_{pji}$  does not depend on  $\kappa_{pj}$  and, therefore, it is considerably easier to implement empirically. In section 3, we will discuss how we approximate  $\xi_p$  and  $\alpha_{pji}$ .

### 2.3 AREER at the Country-Destination Level

The methodology developed to calculate AREER at the country level can also be used to derive countrydestination-level AREERs, such as the Mexican AREER in the US market. These measures can be quite useful for producers, policymakers and researchers alike. Producers may want to focus their export efforts in markets in which they are gaining competitiveness. Conversely, they may want to price to market so as to prevent the temporary loss of a market in which they have lost exchange rate competitiveness (Berman et al., 2012; Burstein and Gopinath, 2015). Policymakers, in turn, may want to take into account destination level AREERs, for example, to focus their export promotion efforts on trading partners in which these efforts are more likely to bear fruit. Researchers may want to use country-destination-level AREERs to study the impact of exchange rate competitiveness on trade using gravity models with bilateral trade data.<sup>16</sup> While all these actors can use traditional bilateral RERs to address these issues, a destination-level AREER offers a more precise measure of a country's competitiveness in each of its markets. Here we just extend these traditional bilateral measures to incorporate competition from third countries. Our measures are "bilateral" in the sense that they refer to the real exchange rate of country i in market j. But they are multilateral in that they take into account the fact that producers in country i exporting to country j are also competing with producers of similar products from other countries. We call these measures destination-level AREERs, or  $AREER_i^j$ .

As a starting point, we define the country-destination-level AREER as follows:

$$AREER_{i}^{j} = \prod_{\substack{k=1\\k\neq i}}^{n} RER_{ik}^{w_{ik}^{j}} = RER_{ij}^{w_{ij}^{j}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ik}^{w_{ik}^{j}},$$
(16)

where  $w_{ij}^{j}$  is the weight of country j in  $AREER_{i}^{j}$  and  $w_{ik}^{j}$  is the weight assigned to country k in the same computation. Since in these destination-level AREERs we are just considering exports to country j, the weight corresponding to the destination country itself will be  $w_{ij}^{j} = \alpha_{ji}$ , that is, the share of absorption of tradables (not originated in i) in j satisfied by domestic producers. Likewise, we specify  $w_{ik}^{j}$  in the following way:

$$w_{ik}^j = (1 - \alpha_{ji}) \frac{M_{jk}}{M_{j,-i}},$$

where the share of absorption of tradables satisfied by imports (not originated in i) or  $(1 - \alpha_{ii})$  is distributed

 $<sup>^{16}</sup>$ Until now, all the existing empirical literature looking at the impact of real exchange rates on trade using bilateral data, reviewed for example by Auboin and Michele (2013), relies on traditional bilateral real exchange rates.

according to the importance of each country k in the destination market j.<sup>17</sup> From these definitions it follows that  $w_{ij}^{j}$  captures the degree of competition between producers in countries i and j in market j, while  $w_{ik}^{j}$  captures the competition of i with each third country k in the destination country j.<sup>18</sup>

Notice that if  $\alpha_{ji} = 1$ , then  $w_{ij}^j = 1$ ,  $w_{ik}^j = 0$ , and  $AREER_i^j$  reduces to  $RER_{ij}$ , which is the traditional bilateral real exchange rate. In other words, if  $\alpha_{ji} = 1$ , producers in country *i* only compete with their counterparts from country *j* in that particular market, and thus the traditional bilateral real exchange rate applies. It can be shown that the country-level AREER can be calculated as the weighted geometrical average of the destination-level AREERs, with the weights assigned to each destination being the traditional real exchange rate weights,  $X_{ij}/X_{i.}$ , that is, the share of each destination country *i*'s exports (see Mathematical Appendix).

Similar to AREERs at the country level, AREERs at the country-destination level can also be modified to take into account similarity of export baskets between the exporting country and its competitors. Furthermore, weights assigned to each competitor in the destination country can be computed with manufacturing data only or as a composite index of the weights for manufactures and commodities. Given that those adjustments resemble those implemented in the AREER at the country level (subsection 2.1), we do not discuss how they are extended to the country-destination case.

### 2.4 AREER at the Country-Product-Destination Level

Following equation (16), it is possible to define an adjusted real effective exchange rate at the countryproduct-destination level. In particular,

$$AREER_{ip}^{j} = RER_{ij}^{w_{pij}^{j}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ik}^{w_{pik}^{j}} = RER_{ij}^{\alpha_{pji}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ik}^{(1-\alpha_{pji})\frac{m_{pjk}}{m_{pj,-i}}}$$
(17)

where  $w_{pij}^{j}$  is the weight that country j receives in country i's exchange rate for product p in market j, thus corresponding to the competition between producers from countries i and j in market j for product p. Conversely,  $w_{pik}^{j}$  captures the competition between producers from country i and each third country k in the product market p of country j. Unlike the aggregate AREER, the AREER at the country-destination-product level does not require adjustments by export similarity since it is assumed that the disaggregation used for p does not allow us to calculate the similarity index defined above.<sup>19</sup> As discussed in the previous section for the case of the country level AREER, it is also possible to calculate the product level  $AREER_{ip}$  as the weighted average of the product-destination level AREERs, using the shares of country i's exports to each destination as weights.

 $<sup>{}^{17}</sup>M_{jk}$  are country j's imports from country k, while  $M_{j-i}$ , are country j's imports from the world, except those from country

*i.* <sup>18</sup>For ease of exposition we have not included in the discussion the adjustment for similarity. However, all results discussed in this article regarding the destination-level AREERs incorporate that additional adjustment, as discussed in Subsection 2.1.2 above.

 $<sup>^{19}</sup>$  If p corresponds to aggregate data at the sector level, however, it is possible to implement a similarity correction similar to the one used at the country level.

### 3 Data

### **3.1 Data Sources**

To compute bilateral real exchange rates, we use monthly average nominal exchange rates and end-of-themonth consumer price indices (CPI) from the International Monetary Funds' International Financial Statistics (IFS). For Latin America and the Caribbean, we complement those cases with missing information with official data sources from national central banks and statistical offices. We use CPI data, instead of other measures such as unit labor cost (ULC) and producer price indices (PPI), due to their availability and standard calculation methodology across countries. For this reason, our AREERs measure price competitiveness in opposition to cost competitiveness (Chinn, 2006).

Bilateral export and import data for the calculation of the weights come from the BACI product-level international trade database, reported by the Centre d'Etudes Prospectives et d'Informations Internationales (CEPII). This database reconciles declarations of exporters and importers in the United Nations Commodity Trade Statistics Database (COMTRADE) (Gaulier and Zignago, 2010). In our computations, we include information for 120 countries and 769 classes of goods (P) (4-digit SITC Rev. 2) for the year 2013. As is standard in the literature, we use export data that excludes exports of foreign goods to ensure that our competitiveness measures are not being driven by reexports. For the calculation of  $\alpha_{ji}$ 's, however, trade data needs to be comparable to our measure of aggregate local production, which comes from the World Development Indicators (WDI). Since there are discrepancies between aggregate trade data from WDI and BACI, we weight bilateral exports and imports from this last source in order to match the aggregate exports and imports reported in WDI.<sup>20</sup>

Regarding production, the value added of tradables at the country level is approximated with industrial and agricultural GDP from WDI for the year 2013  $(Y_j \text{ in } (3))$ . To construct country-product-specific output data  $(y_{pj}^* \text{ in } (15))$ , we rely on data from the United Nations Industrial Development Organization (UNIDO), which contains information on value added and on gross output at the 4-digit International Standard Industrial Classification (ISIC) Rev. 3 level for the period 1990 onward. To match the production information with trade data at 4-digit SITC Rev. 2 for the calculation of the  $\alpha_{pii}$ , we use the correspondence table between ISIC Rev. 3 and 4-digit SITC Rev. 3, available in Eurostat, and the crosswalk between SITC Rev. 3 and SITC Rev. 2. In cases in which there are multiple ISIC categories that match a single SITC code, production is simply aggregated. In cases in which a single ISIC production activity matches multiple SITC trade categories, we distribute the production among the corresponding trade categories according to the proportion of exports of each product within the ISIC activity in  $2013.^{21}$  Given that UNIDO has significant differences in product and time coverage across countries, when the production information corresponding to the year 2013 is not available, we use the last year with available information for each country-product pair.<sup>22</sup> Additionally, we complement UNIDO data with information on production of agricultural commodities from the Food and Agriculture Organization of the United Nations' Statistics Division (FAOSTAT).<sup>23</sup> To measure  $\xi_p$ , we compute the ratio between the value added and the gross output for all the countries and products with available information.  $\xi_p$  is thus approximated as the average for each p of these ratios.

<sup>&</sup>lt;sup>20</sup>Specifically, we multiply the bilateral exports from BACI by the factor  $\frac{X_i}{\sum_j X_{ij}}$ , where  $X_{ij}$  are the exports from *i* to *j* as reported in BACI and  $X_i$  are *i*'s total exports from WDI.

<sup>&</sup>lt;sup>21</sup>In this last case, the implicit assumption is that the share of each product in production is similar to its share in exports. <sup>22</sup>For consistency, in such cases we use trade data for the same year for the calculation of the  $\alpha_{pji}$ .

 $<sup>^{23}</sup>$ Since production and trade data of agricultural commodities is highly disaggregated, matching the FAOSTAT production data with the SITC Rev. 2 trade information does not pose further challenges.

Information on the local value added in exports at the country and country-sector (2-digit GTAP classification) level ( $\tau_j$  in (3)) comes from Blyde (2014), based on Purdue University's Global Trade Analysis Project database for the year 2007 (Aguiar et al., 2016). At the country level, we complement the data using information from the OECD-WTO Trade in Value Added (TiVA) database also for the year 2007 (OECD, 2013). To construct a database of value-added exports at the 4-digit SITC Rev. 2 level, we use the crosswalks available in World Integrated Trade Solutions (WITS) between GTAP and HS2 and, subsequently, between HS2 and SITC Rev. 2. In those cases where more than one value-added exports ratio (VAX) at the sector level maps to one SITC category, we compute a simple average to approximate the VAX for that product. In total, we match 43 different GTAP sectors to our product-level export data.

Computing country-product-level AREERs for a given country *i* and product *p* requires data on all  $\alpha_{pji}$  corresponding to that country and product.<sup>24</sup> This in turn requires information on local production of product *p* in every other country. When this information is missing,  $\alpha_{pji}$  is missing, and we need to impute it.<sup>25</sup> To do this, we use boosted regression, as proposed by Hastie et al. (2009) and James et al. (2013). The imputation method is discussed in detail in Appendix B.

### **3.2** Data Dissemination

We use the methodology and data sources described above to build three novel datasets for competitionand similarity-adjusted REERs at the country, country-product and country-destination level.<sup>26</sup> These datasets are publicly available.<sup>27</sup> All datasets include AREERs that treat manufactures and commodities symmetrically (i.e, AREER). Moreover, at the country and country-destination level we provide the AREER-M, which uses only manufacturing trade flows, and the AREER-MC, which combines different weights for trade in manufactures and commodities into a single composite index. Furthermore, for ease of comparison we provide bilateral exchange rates (real and nominal) and traditional real effective exchange rates at the country and country-product level. All databases have information for 120 countries and exhibit monthly periodicity, spanning the period between January 2014 and (at the time of this writing) July 2017. We plan to update the time coverage periodically as monthly CPI and exchange rate data become available. Moreover, all three databases use fixed weights based on trade data for 2013.<sup>28</sup> The dataset at the country-product level provides information for 769 products at 4-digit SITC Rev. 2 classification.

### 4 A Case Study: Latin America and the Caribbean during a Period of Strong Exchange Rate Movements

In this section, we focus on the period between May 2014 and February 2016, characterized by strong nominal and real exchange rate movements in most of Latin America and the Caribbean as well as in the

 $<sup>^{24}</sup>$  For example, computing Mexico's AREER for cars requires information on the local share in the absorption of cars in every other trading partner.

<sup>&</sup>lt;sup>25</sup>While the number of missing observations for domestic production is very large, they tend to correspond to small countries that have correspondingly small weight in country's export baskets. In the case of manufactures, countries with missing production data account on average for 26 percent of exports in the entire sample.

<sup>&</sup>lt;sup>26</sup>Note that because of its high dimensionality, we only computed the dataset at the country-product-destination level for some products, including our example of color TVs (SITC Rev. 2 category 7611) discussed below. Specific data at the country-product-destination level are available upon request.

 $<sup>^{27}</sup>$  The data can be found at https://sites.google.com/site/andresfernandezmartin8/research.

 $<sup>^{28}</sup>$  While using fixed weights is appropriate for the small time period covered so far, eventually we plan to update the weights using the Laspeyres chained-linked methodology.

United States, in order to illustrate the value of our adjusted REER measures.<sup>29</sup> Figure 2 presents the change in real exchange rates during this period. Fifteen out of the 23 countries included in our Latin America and the Caribbean sample experienced real depreciation, although the size of the depreciation varies substantially from country to country. In many cases, depreciation was very substantial, particularly in Colombia, Brazil, and Mexico. In contrast, five countries experienced real appreciation. During this same period, the currency of Latin America and Caribbean countries' main trading partner, the US dollar, appreciated by 21.2 percent percent, using the traditional REER. The diverse exchange rate experience in Latin America and the Caribbean as well as in the rest of the world generates the potential for the changes in weights introduced by any alternative methodology to lead to noticeable differences in exchange rate competitiveness, in comparison to those that would emerge by using traditional REERs. One of the key drivers of these differences is precisely the comparatively lower weight of the United States in the computation of the AREERs of Latin America and the Caribbean. In what follows, we illustrate these issues by analyzing exchange rate weights and patterns at the country, country-destination, country-product, and country-product-destination levels.

[Figure 2 about here.]

### 4.1 AREER at the Country Level

Before analyzing the evolution of real exchange rates, it is useful to look at the changes in weights associated with the adjustments for competition in third markets and export basket similarity. Figure 3 compares how weights used in the computation of the AREER and REER differ for select countries.<sup>30</sup> Panel a) reports the results for Mexico. In line with the examples provided throughout the article, the traditional REER weight for the United States is 71.4 percent, as the United States is the destination of 71.4 percent of Mexican total exports. Once we take into account competition in third markets and export basket similarity, however, the weight corresponding to the United States in Mexico's AREER declines to 47.5 percent, a decline of 24 percentage points. Conversely, China's weight increases from 2.7 percent to 11 percent, since it is not an important destination of Mexican exports, but competes with Mexican exporters in other markets, and in similar products. Notice that most countries in the Figure for Mexico gain weight in the adjusted index, compensating for the huge loss corresponding to the United States. The only exception among the 10 most important countries presented in the Figure is Spain, which is more important as an export destination than as a competitor in third markets. Panels b) and c) in the Figure suggest that the pattern is more or less similar in other countries, such as Colombia and Ecuador.<sup>31</sup> In this last case, the weight of the United States is cut by two-thirds, which, as we will see below, has an important impact on the evolution of the Ecuadorean real exchange rate.

### [Figure 3 about here.]

<sup>&</sup>lt;sup>29</sup>This period maximizes the change in real bilateral exchange rates of Latin American and Caribbean countries vis-à-vis the dollar, as well as the appreciation of the dollar as measured by the traditional REER.

<sup>&</sup>lt;sup>30</sup>The AREER we used for this figure is the first one discussed in Section 2, that is, the one that treats manufactures and commodities symmetrically. Corresponding figures for the other two versions of our aggregate index (AREER-M and AREER-MC) yield qualitatively similar results, and are available upon request.

 $<sup>^{31}</sup>$ Colombia is the country with the largest real depreciation in the region. Ecuador, which is dollarized, experienced a large real appreciation and, as we will see below, is the country that exhibits the largest difference when using AREERs rather than REERs.

Figure 4 summarizes the change in the structure of weights between the AREER and REER indices across Latin America and Caribbean countries (see also Table A1 in the Appendix). The bars, which are symmetric around the origin (since any country's weight gain is compensated by another country's weight loss), show the extent of reallocation of weights for each country, under each of our three measures of aggregate AREERs, vis-à-vis traditional REERs. Panel a) shows that the reallocation for Mexico (28 percentage points) is more or less typical in the degree to which weights shift when switching from traditional weights to AREER weights. As shown in Table A1 in the Appendix (column 1), the shift in weights for countries in Latin America and the Caribbean average 31.2 percent, while the corresponding figure for the world sample is 32.8 percent. These are major shifts in weights, which can potentially lead to important differences in the pattern of real exchange rates.

Figure 4 also shows that there is a great deal of variation in the magnitude of weight changes within Latin America and the Caribbean. Shifts in weights range from 15.6 percent in Bolivia to 48.5 percent in Haiti. These shifts are even greater for our AREER-M and AREER-MC measures, ranging from 24.6 (Costa Rica) to 60.7 percent (Suriname) in AREER-M, and from 26.5 (Costa Rica) to 73.7 percent (Suriname) in AREER-MC.<sup>32</sup> Average shifts in wheight are similar for the world sample as a whole. The blue and red bars show the weight gains and losses for the United States and China, respectively. On average, the US weight losses amount to 10.5 percentage points (AREER), 7.8 percentage points (AREER-MC). China, in contrast, gains on average 3.6, 6.9 and 0.4 percentage points in weight, respectively. It is interesting that, while China gains considerable weight in countries such as Mexico with which they compete in third markets, it loses substantial weight in countries such as Brazil, Chile and Uruguay, for which it represents an important export destination.

### [Figure 4 about here.]

How do these changes in weights translate into differential exchange rate patterns? Figure 5 shows the changes in the REER (in blue in the Figure) and the different versions of the AREER (in red).<sup>33</sup> Thus, they illustrate how the changes in the weights discussed above influenced export competitiveness in the period between May 2014 and February 2016. Notice that while most countries experienced real bilateral depreciation vis-à-vis the US dollar during this period (Figure 2), the majority of them actually exhibited substantial real effective appreciation, leading to losses in export competitiveness. Only three countries—Colombia, Brazil and Mexico—experienced substantial real depreciation, regardless of how it is measured. There are, however, differences between the behavior of REER and our AREER measures. In particular, on average, economies in Latin America and the Caribbean appreciated more markedly than the traditional REER suggests. According to our first measure (AREER), depicted in panel a), Latin American and Caribbean currencies appreciated on average 7.9 percent from May 2014 to February 2016; the corresponding figure for the REER is only 5.1 percent. In fact, each country in the figure, with the exception of Paraguay, loses relative competitiveness when using AREER rather than REER, either because the depreciation is smaller, or the appreciation larger. At the other extreme, for Ecuador, appreciation increases from 20.5 percent under REER to 33.2 percent under AREER, a very significant change that can have a huge impact on the ability of firms to compete. Even the average change from 5.1 percent appreciation

<sup>&</sup>lt;sup>32</sup>The greater shift in weights for AREER-MC may be linked to the fact that, for the commodity component of this index that is important for large commodity exporters in the region, weights shift from commodity importers (in the traditional REER) to commodity exporters, since it is with the latter that Latin America and Caribbean countries are competing in the world market that is relevant for the AREER-MC computation.

<sup>&</sup>lt;sup>33</sup>Table A2 in the Appendix shows the change in the competitiveness from May 2014 to February 2016 when comparing the REER with the different versions of the AREER.

to 7.9 percent appreciation may have a significant impact on the ability of firms to compete when markets are very competitive and markups are small. While the differences are smaller in the case of our measure focused on manufactures (AREER-M), they are largest in the case of AREER-MC, where the average appreciation increases from 5.1 to 9.1 percent.

### [Figure 5 about here.]

Figure 6 captures the evolution of our first AREER index during the period under study, in solid red, for Mexico, Colombia and Ecuador, the same countries for which we showed the shift in weights in Figure 4 above. For comparison, the plots also report the evolution of the nominal exchange rate against the US dollar (solid black), the bilateral real exchange rate with the United States (dashed black) and the traditional REER without adjusting for competition or similarity (solid blue). All four variables are normalized to 100 at the beginning of the period of analysis. Notice that in the three cases shown, as we discussed above, the AREERs are always higher (i.e., appreciate more or depreciate less) than the traditional REER's. This is the case for most countries in Latin America and the Caribbean, as the weight of the US dollar -a currency that has appreciated substantially over this period- tends to decrease with the competition and similarity adjustment. The additional loss of competitiveness due to the adjustment ranges in these three countries from 2.8 percent in Mexico to 12.7 percent in Ecuador.

[Figure 6 about here.]

### 4.2 AREER at the Country-Destination Level

Figure 7 illustrates the change in the weights used in the construction of the country-destination-level AREER, that is, the real effective exchange rates of a country in a specific destination. For illustration purposes, we focus on the United States as a destination for Mexico and on Canada as a destination for Argentina. Traditional bilateral real exchange rates (RER), which would be the obvious traditional alternative to our country-destination-level AREERs, would assign a weight equal to 100 percent to the destination country, since third market competition would not be considered. Each bar in the figures shows the weight assigned to the destination and third-market competitors. Using the country-destination-level AREER reduces the weight of the United States from 100 to 58.5 percent, in the case of Mexico, reassigning these weights to Mexico's most important competitors in the US market, mainly China and Canada. In the case of Argentina, Canada's weight as a destination is reduced from 100 percent to 37.8 percent, and reassigned mainly to the United States, Argentina's main competitor in the Canadian market. Notice that the United States is a more important competitor in Canada than Canada itself, and thus has a slightly larger weight. How do these destination country weights look more generally? In fact, for the world sample as a whole, the average weight of the destination country in its own market (that is, the average  $\alpha_{ji}$ ) is 37.2 percent for the AREER-M and 22.8 percent for the AREER-MC, as shown in Table 2.

To see how these changes in the weights translate into the AREER at the country-destination-level, consider Figure 8. Each panel shows the change in the country-destination-level AREER between May 2014 - February 2016 for the ten most important export destinations (in order) for Mexico (a) and Argentina (b). Vertical red lines correspond to the change in AREER at the country level (for Mexico and Argentina, respectively). In panel a), the country-destination-level AREER depreciated less than the bilateral RER in some of Mexico's most important destination markets, notably in the United States, China and India.

Since the RER only reflects competition from producers in destination markets, this implies that Mexico's third-country competitors in these destination markets experienced depreciations as well, counteracting competitiveness-enhancing effects stemming from Mexico's own depreciation during that period. In contrast, Mexican producers barely lost competitiveness in Brazil and Colombia, despite the significantly larger depreciation in these two countries, as part of the weight shifted from these destinations to other competitors. In the case of Argentina (panel b), Canada offers an interesting illustration: while using bilateral RER would point to a loss of competitiveness of 7.6 percent, the destination AREER shows that Argentina has a gain of competitiveness of 2.7 percent in the Canadian market, as a large share of weight is reassigned to the United States.<sup>34</sup>

[Figure 7 about here.]

[Figure 8 about here.]

[Table 2 about here.]

### 4.3 AREER at the Country-Product Level

To illustrate the relevance of country-product-level AREERs, we will focus on Argentina. Figure 9 shows the shift in weights for the product-level AREER vis-à-vis the product-level REER for the 15 most important export products in this country, according to their share in 2013 total exports. As can be seen, taking into account third market competition changes significantly the structure of weights.<sup>35</sup> Concretely, the average weight shift is 61 percent for the products reported in Figure 9 and 63 percent for the 714 products exported by Argentina and included in the computations. However, these averages hide important across-product variation, with categories such as chemicals and trucks and vans showing shifts in their weights of less than 30 percent, and copper ores and gold shifting more than 90 percent of the weight.

### [Figure 9 about here.]

How do these changes in the structure of weights translate into changes in competitiveness? Figure 10 summarizes the change in the product-level REER and AREER for the period May 2014-February 2016 for Argentina. Each bubble corresponds to one 4-digit SITC product, with the size of each bubble being proportional to the share of the product in the country's exports for the 15 products reported in Figure 9. Two findings stand out: first, there is huge variation in the evolution of the AREER at the product level, with most products, including all the most important ones, showing changes in the AREER between -10 and 20 percent.<sup>36</sup> This shows that, while most products lost competitiveness during this period, about one third of the products actually became more competitive. Second, the difference in the structure of weights between the REER and AREER at the product level leads to important differences in the evolution of the exchange rate competitiveness when measured with product-level AREERs, rather than product-level REERs. An extreme example is that of barley, a product with a relatively high shift in weights, which experienced a loss

<sup>&</sup>lt;sup>34</sup>Not surprisingly, panel a) shows that Mexico also achieved higher competitiveness in Canada, for similar reasons.

<sup>&</sup>lt;sup>35</sup>Recall from section 4.3 that adjustment for export basket similarity is not relevant at the product level.

<sup>&</sup>lt;sup>36</sup>Among the most important products, the one that lost the most competitiveness is cars, particularly when using REER, since 95 percent of exports go to Brazil, a country that experienced substantial depreciation. The loss of competitiveness is somewhat smaller with AREER, since Argentina also competes in the Brazilian market with other countries.

of competitiveness according to the AREER measure, but gains according to REER. This can be explained by a redistribution in the weights from Saudi Arabia (the main destination for the Argentinian exports of barley and a country that experienced a real appreciation against the United States) to Ukraine and Rusia (the main competitors in the Saudi barley market, both countries with the largest real depreciations during the period under consideration). In contrast, a product like cars, with a relativley low shift in weights, is much closer to the 45 degree line. While we chose to focus on Argentina for illustrative purposes only, the standard deviations reported in Table 3, column 3 show that the dispersion of product-level AREERs in this country is not an exception, but rather close to the norm.

[Figure 10 about here.]

[Table 3 about here.]

### 4.4 AREER at the Country-Product-Destination Level

Recall that, in the introduction, we motivated the need to adjust for third-country competition in the computation of real effective exchange rates using the example of Mexican color TV producers competing with Chinese producers in the US market. To end this case study, we come back full circle to this example as a way to discuss the relevance of our country-product-destination level AREERs. Figure 11 shows the composition of the absorption of color TVs in the United States by country of origin. As we can see in the figure, Mexico (depicted in yellow) is the main supplier, followed closely by China (depicted in brown). Out of the non-Mexican TVs, the United States (in blue) is only responsible for 3.3 percent of absorption. So this is the weight of the United States in the country-product-destination level AREER corresponding to Mexican TV's in the United States. As can be seen in Figure 12, by far the largest weight corresponds to China, which is responsible for supplying 73 percent of non-Mexican color TVs to the US market. <sup>37</sup> Accounting for China, and other suppliers, significantly impacts the competitiveness of Mexican TVs in the US market. As seen in panel b), while competitiveness after accounting for third market competition is just 19.9 percent. The main reason for the discrepancy is that, during the same period, the bilateral real exchange rate of China vis-à-vis the United States also depreciated, in this case by 6.4 percent.

[Figure 11 about here.]

[Figure 12 about here.]

### 5 Concluding Remarks

In this paper, we develop new measures of real effective exchange rates that consider competition in third markets, as well as similarity of export baskets between countries, in the definition of the relevant weights. In addition to competition-adjusted real effective exchange rates at the aggregate country level, we extend

 $<sup>^{37}</sup>$ It is worth pointing out that Korean companies such as Samsung and LG have production facilities in Mexico, from which they export to the United States.

the methodology to develop measures at the country-product, country-destination, and country-productdestination levels. To the best of our knowledge, none of these more disaggregated competition-adjusted REERs have been developed before.

We build a new dataset where we apply this methodology to compute monthly adjusted REERs at the country, country-product, and country-destination levels using data on exchange rates, prices, production, and bilateral trade for 120 countries and 769 products at the 4-digit SITC level (Revision 2). We make this dataset publicly available. Lastly, to illustrate the usefulness of our methodological contribution and our dataset, we show the impact of these adjustments on the exchange rate weights and patterns for countries in Latin America and the Caribbean between May 2014 and February 2016 at the country, country-product, country-destination, and country-product-destination levels. This is a period characterized by substantial movement in both nominal and real exchange rates in the countries in the Latin American and Caribbean region, as well as a large real appreciation in the United States.

Our results for aggregate exchange rates show that the weights for the adjusted AREER measure are significantly different from the traditional REER weights that leave out competition in third markets, leading to important differences in the evolution of countries' export competitiveness. On average, between 31 and 49 percent of the weights corresponding to Latin America and the Caribbean countries shift as a result of the adjustment, depending on the AREER used. For the world sample, the average shift in weight ranges between 32 and 44 percent, depending on the AREER specification. This means, that, by leaving aside competition in third markets and export basket similarity, traditional measures of the real effective exchange rate misallocate between one third and one half of the weights. A country like the United States, which is important as an export destination but not so relevant as competitor in third markets, loses on average between 8 and 16 percentage points in weight, depending on the specification. Meanwhile, China gains significant weight for countries such as Mexico and most of Central America, where it is an important competitor, but loses weight in countries such as Brazil, Chile, and Uruguay, for which it is an important export destination.

This shift in the relevant weights affects the evolution of countries' exchange rate competitiveness during the period under study. In general, accounting for competition in third markets and similarity led to larger losses of competitiveness. While real appreciation for countries in Latin America and the Caribbean during this period amounted to 5.1 percent on average using traditional REER, the loss of competitiveness ranges between 6.3 and 9.1 percent using our measures, depending on the specification. While these additional losses vary from country to country, 18 out of 23 countries in Latin America and the Caribbean experience diminished competitiveness vis-à-vis traditional measures, regardless of specification.

At the country-destination level, using our competition adjusted measures reduces the weight of the destination country from 100 percent (as in traditional bilateral RERs) to an average ranging between 23 and 45 percent, depending on the specification. The shift in weights significantly alters the pattern of exchange rate competitiveness in specific destination markets. A case in point is that of Argentina's relevant AREER vis-à-vis Canada. While using bilateral RERs which allocate 100 percent of the weight to the destination country would result in a loss of competitiveness of 8 percent, Argentina actually gained competitiveness in the Canadian market according to our destination-level AREER, as much of the weight shifts from Canada to the United States.

Our results also show that products within countries exhibit very different experiences in terms of their exchange rate competitiveness. The reason is that the structure of weights, which depends on which countries a given country trades with in each product and with whom it competes in those markets, differs significantly

across products. Thus, while aggregate shocks that affect a country's exchange rates may reduce the competitiveness of some products, the same shocks may enhance the competitiveness of others. For example, in Argentina, most products experienced changes in AREER between -10 and 20 percent, with a standard deviation of 7 percentage points. These dissimilar experiences justify our efforts to develop competition-adjusted REERs at the country-product and the country-product-destination levels.

While we think our paper makes significant contributions to the measurement of real effective exchange rates, it does have shortcomings. One of them is that, unlike recent work by Bems and Johnson (2017), Bayoumi et al. (2013), and Patel et al. (2014), it does not account for trade in value added, which would be desirable given the increasing role of global value chains in world trade. This is an important consideration, which accounts for the fact that imports from one country could contain inputs produced in others. This is not captured in our measures, which assign weights according to the countries from which products are directly imported. At the time of this writing, however, the input-output data necessary to develop measures that account for trade in value added have limited country coverage, particularly in developing countries which are the focus of our interest. We hope to be able to address this limitation in the future, as more data become available.

Another shortcoming is that our analysis does not take into account trade in services. This is an important limitation, in particular for countries such as those in the Caribbean, in which services comprise an important share of exports. Unfortunately, while there are some data on bilateral trade in services, the quality of these data is questionable. Finally, while we work with CPI inflation data, at least for AREERs at the product level, it would be more appropriate to work with disaggregated price (or cost) data, relevant for each of the products considered. Once again, our choice in this case is driven by data availability. In spite of these shortcomings, we believe that the benefits of introducing our AREER measures for such a wide range of countries far outweigh the costs associated with those limitations.

Our competition-adjusted exchange rates can be useful for policymakers as well as researchers. At the aggregate level, our measures provide policymakers with a tool to track exchange rate competitiveness more accurately by accounting for exchange rate movements not just vis-à-vis trading partners, but also countries with which they compete in third markets. This may help understand why aggregate exports may not be responding according to expectations, if expectations are built on the basis of traditional measures. Relatedly, from a macroeconomic perspective, using our measures may provide more accurate expectations on the speed of external adjustment of the current account. In the period under consideration, thinking in terms of bilateral RERs (which depreciated substantially for most countries in the region) led to undue expectations of quicker adjustment, when in fact the AREERs were in most cases actually appreciating.

At the more disaggregated level, information on real exchange rates at the product and product-destination level may be a useful tool in helping to guide export promotion policies. Such policies may help firms maintain markets where competitiveness has declined, or alternatively to help break into new markets in products that have gained competitiveness. Focusing promotion efforts on the latter may offer more "bang for the buck" for those in charge of export promotion policies. In addition, these measures may help policymakers establish which export sectors are suffering and may benefit from support. For instance, loss of exchange rate competitiveness could potentially be used as a criteria for policies such as the US Trade Adjustment Assistance program.

Researchers should also benefit from having more accurate aggregate measures of exchange rate competitiveness, as well as measures of AREER at the country-destination, country-product, and country-productdestination levels. For example, our country-destination level measures can be used to study the impact of real exchange rates on exports within the context of a gravity model, in place of the traditional bilateral RER measures that are a less accurate characterization of the competitiveness of a country's exports in specific destinations.

### References

- Aguiar, A., B. Narayanan, and R. McDougall (2016). An overview of the gtap 9 data base. Journal of Global Economic Analysis 1(1), 181–208.
- Armington, P. (1969). A theory of demand for products distinguished by place of production. IMF Staff Papers 16, 159–178.
- Artus, J. and A. McGuirk (1981). A revised version of the multilateral exchange rate model. *IMF Staff* Papers 28.
- Artus, J. and R. Rhomberg (1973). A multilateral exchange rate model. IMF Staff Papers 20.
- Auboin, M. and R. Michele (2013). The relationship between exchange rates and international trade: a literature review. *World Trade Review* 12(3), 577–605.
- Bayoumi, T., J. Lee, and S. Jayanthi (2006). New rates from new weights. *IMF Staff Papers* 53(2).
- Bayoumi, T., M. Saito, and J. Turunen (2013). Measuring competitiveness: Trade in goods or tasks? IMF Working Papers 13(100).
- Bems, R. and R. C. Johnson (2017). Demand for value added and value-added exchange rates. *American Economic Journal: Macroeconomics (Forthcoming)*.
- Bennett, H. and Z. Zarnic (2009). International competitiveness of the mediterranean quartet: A heterogeneous-product approach. *IMF Staff Papers* 56 (4), 919–957.
- Berman, N., P. Martin, and T. Mayer (2012, jan). How do different exporters react to exchange rate changes? The Quarterly Journal of Economics 127(1), 437–492.
- Blyde, J. (2014). Synchronized factories: Latin America and the Caribbean in the era of global value chains. Springer Open.
- Boughton, J. (1997). Modeling the world economic outlook at the IMF: A historical review. *IMF Working* Paper 48.
- Buldorini, L. (2002). The effective exchange rates of the Euro. ECB Occasional Paper Series 2.
- Burstein, A. and G. Gopinath (2015). *Handbook of International Economics*, Volume 4, Chapter International Prices and Exchange Rates, pp. 391–451. Elsevier B.V.
- Chinn, M. (2006). A primer on real effective exchange rates: Determinants, overvaluation, trade flows and competitive devaluation. Open Economies Review 17, 115–143.
- Finger, J. and M. Kreinin (1979, dec). A measure of 'export similarity' and its possible uses. The Economic Journal 89(356), 905–12.
- Gaulier, G. and S. Zignago (2010). BACI: International trade database at the product-level, the 1994-2007 version. *CEPII Working Paper 23*.
- Hastie, T., R. Tibshirani, and J. Friedmand (2009). The elements of statistical learning: Data mining, inference, and prediction (Second edition ed.). Springer.

- Hausmann, R., C. Hidalgo, S. Bustos, M. Coscia, S. Chung, A. Simoes, and M. Yildirim (2013). *The atlas of economic complexity: mapping paths to prosperity.* MIT Press.
- James, G., D. Witten, T. Hastie, and R. Tibshirani (2013). An introduction to statistical learning with applications in R. Springer Texts in Statistics. Springer.
- Klau, M. (2006). The new BIS effective exchange rate indices. BIS Quaterly Review (March), 51-66.
- McGuirk, A. (1986). Measuring price competitiveness for industrial country trade in manufactures. *IMF Working Paper 34*.
- OECD (2013). Interconnected economies: Benefiting from global value chains. OECD Publishing.
- Patel, N., Z. Wang, and S.-J. Wei (2014). Global value chains and effective exchange rates at the countrysector level. *NBER Working June*(20236).
- Rhomberg, R. (1976). Indices of effective exchange rates. IMF Staff Papers 23.
- Schonlau, M. (2005). Boosted regression (boosting): An introductory tutorial and a Stata plugin. The Stata Journal 5 (3), 330–354.
- Timmer, M., E. Dietzenbacher, B. Los, R. Stehrer, and G. de Vries (2015). An illustrated user guide to the World Input-Output Database: the case of global automotive production. *Review of International Economics* 23(3), 575-605.
- Zanello, A. and D. Desruelle (1997). A primer on the IMF's information notice system. *IMF Working* Paper 71.

### Figure 1: Calculating $\alpha_{US,MEX}$ and $\alpha_{US,COL}$ a. Mexico





Source: Authors' calculations based on: Bilateral trade flows for 2013 from BACI database (Gaulier and Zignago, 2010), as reflected in Hausmann et al. (2013); industrial and agricultural GDP from the World Bank's World Development Indicators; and local value added in exports from Blyde (2014).

Figure 2: Change in Bilateral Real Exchange Rates in Latin America and the Caribbean (May 2014-February 2016)



Note: Each bar shows the percentage change in the real exchange rates against the US dollar from May 2014 to February 2016.



Figure 3: Shift in Weights in Selected Countries: AREER vs. REER

Note: Each bar shows the difference in the weight assigned to each trading partner for the traditional REER and the AREER.



Figure 4: Shift in Weights: AREER vs. REER

Note: Each bar shows the differences in the weights assigned to competitors in the AREER and the REER, with positive (negative) differences meaning that the weight assigned to that country is bigger (smaller) in the AREER than in the REER. The bars are symmetric around the origin since any country's weight gain is compensated by another country's weight loss.



Figure 5: Change in the AREER and REER in Latin America and the Caribbean: May 2014 - February 2016

Note: Each bar shows the percentage change in the exchange rates from May 2014 to February 2016. Vertical lines depict simple averages across countries.



Figure 6: AREER over Time, May 2014 - February 2016



Figure 7: AREER Weights for Mexico and Argentina in Selected Destination Countries

Note: Each bar shows the weight assigned to the destination country vis-à-vis the third-market competitors in the destinationlevel AREER. The bilateral RER uses a weight of 1 for the destination market.



Figure 8: Change in the Country-Destination-Level AREER: 10 Most Important Destinations, May 2014 - February 2016

Note: Each panel shows the change in the destination-level AREER between May 2014 and February 2016 for the 10 most important destination of the exports of Mexico (panel a) and Argentina (panel b). Vertical red lines correspond to the change in the country-level AREER.



Figure 9: Shift in Weights: Product-Level AREER and REER in Argentina

Note: Each bar shows the change in the structure of weights between the AREER and REER at the product level for the 15 most important products in Argentina, according to its export share in 2013 (in parenthesis).



Figure 10: Change in the AREER and REER at the Product Level in Argentina: May 2014 - February 2016

Note: Each bubble corresponds to a 4-digit SITC Rev.2 category. The size of the bubbles for the 15 products reported in Figure 9 is proportional to the share of the product in the 2013 Argentinian exports. For the rest of the products the size of the bubbles was fixed.  $45^{\circ}$  degree line is in red.



Source: Authors' calculations based on: Bilateral trade flows for 2013 from BACI database (Gaulier and Zignago, 2010), as reflected in Hausmann et al. (2013); and production data from UNIDO.



Figure 12: AREER at the Product-Destination Level: Mexican Color TVs in the United States

Note: Panel (a) shows the weights used in the computation of the AREER for Mexican TVs in the US market. Panel (b) shows the evolution of the AREER at the product-destination level and the bilateral RER (dollars per Mexican Peso) for the period May 2014-February 2016.

Table 1: Calculating Competition-Adjusted Weights for Mexico

|  | k                    | USA  | CAN  | CHN  | ESP        | •••   | j's weight coming from: |               |                   |
|--|----------------------|------|------|------|------------|-------|-------------------------|---------------|-------------------|
|  | j 🔪                  | 71.4 | 6.5  | 2.7  | <b>2.0</b> | • • • | j's market              | Third markets | World             |
| j's share in<br>k's tradable<br>demand | USA                  | 54.0 | 36.9 | 2.8  | 2.8        | • • • | 38.6                    | 3.7           | $\overline{42.3}$ |
|  | CAN                  | 8.1  | 35.6 | 0.5  | 0.3        | • • • | 2.3                     | 5.9           | 8.2               |
|  | CHN                  | 11.5 | 8    | 73.7 | 5          | •••   | 2                       | 10.2          | 12.2              |
|  | $\operatorname{ESP}$ | 0.3  | 0.2  | 0.1  | 34.1       | •••   | 0.7                     | 0.4           | 1.1               |
|  | :                    | :    | :    | :    | :          |       | •                       | •             | :                 |
|  | •                    | · ·  | •    | •    |            |       | •                       | •             |                   |

Note: Countries k in the top row are the four most important destinations for Mexican exports. Numbers immediately below represent the share of each of these countries in Mexico's total exports. Countries j in the first column represent Mexico's competitors in third markets. Percentages in the cells when j = k are  $\alpha_{j,MEX}$ , the share of j's absorption of (non-Mexican) tradables satisfied by domestic producers. Percentages in the cells when  $j \neq k$  represent  $(1 - \alpha_{j,MEX}) w_{MEX,j}^{tr}$ . The last three columns show the weights capturing Mexico's direct competition with j's domestic producers  $(\alpha_{j,MEX} \cdot w_{MEX,j}^{tr})$ , the weights capturing Mexico's competition with j in third countries  $(\sum_{\substack{k=1 \ k \neq i,j}}^{n} (1 - \alpha_{ki}) w_{ik}^{tr} \frac{M_{kj}}{M_{k,-i}})$ , and the competition-adjusted weights for Mexico  $(w_{MEX,j}^1)$ .

|                     | (1)         | (2)     | (3)         |
|---------------------|-------------|---------|-------------|
| Exporter country    | AREER       | AREER-M | AREER-MC    |
| Argentina           | 48.9        | 46.5    | 18.8        |
| Barbados            | 36.1        | 37.0    | 23.0        |
| Bolivia             | 63.4        | 55.4    | 5.3         |
| Brazil              | 47.1        | 48.6    | 18.6        |
| Chile               | 51.4        | 55.9    | 9.0         |
| Colombia            | 38.0        | 48.1    | 9.4         |
| Costa Rica          | 43.9        | 46.9    | <b>39.0</b> |
| Dominican Republic  | 42.6        | 44.7    | 26.7        |
| Ecuador             | 38.3        | 51.2    | 4.1         |
| Guatemala           | 45.6        | 45.6    | 18.6        |
| Haiti               | 41.2        | 34.1    | 32.4        |
| Honduras            | 46.1        | 47.9    | 29.5        |
| Jamaica             | 36.5        | 44.0    | 9.2         |
| Mexico              | 53.4        | 55.8    | 43.1        |
| Nicaragua           | 45.7        | 41.9    | 25.7        |
| Panama              | <b>39.0</b> | 40.1    | 24.5        |
| Paraguay            | 52.3        | 47.7    | 10.3        |
| Peru                | 43.9        | 51.3    | 8.5         |
| Salvador            | 51.4        | 50.9    | 38.5        |
| Suriname            | 35.3        | 37.9    | 4.9         |
| The Bahamas         | 35.8        | 45.0    | 31.1        |
| Trinidad and Tobago | 40.6        | 52.0    | 22.1        |
| Uruguay             | 51.4        | 51.1    | 16.9        |
| Average LAC         | 44.7        | 46.9    | 20.4        |
| Average RoW         | 35.5        | 44.5    | 23.4        |
| Average World       | 37.2        | 45.0    | 22.8        |

Table 2: Average Weight Assigned to the Destination Country

Note: Column (1) shows the average weight assigned to the destination in the exporter country's destination-level AREERs, where each destination is weighted according to its share in exports. Column (2) reports the same quantity when the AREER is computed with only manufacturing goods and column (3) when commodities are included following the method proposed by Bayoumi et al. (2006). The last three rows show the simple average of the figures at the country level.

|                     | (1)               | (2)                   | (3)                 | (4)   | (5)   |
|---------------------|-------------------|-----------------------|---------------------|-------|-------|
|                     | $\mathrm{mean}^*$ | $\operatorname{mean}$ | $\operatorname{sd}$ | p25   | p75   |
| Argentina           | 7.1               | 2.8                   | 7.0                 | -1.9  | 5.9   |
| Barbados            | 11.6              | 9.1                   | 7.2                 | 4.7   | 12.7  |
| Bolivia             | 17.8              | 24.6                  | 8.5                 | 18.7  | 28.4  |
| Brazil              | -23.2             | -24.7                 | 4.3                 | -27.0 | -23.3 |
| Chile               | 1.8               | -1.5                  | 5.8                 | -5.2  | 1.3   |
| Colombia            | -29.3             | -29.6                 | 4.2                 | -32.2 | -27.6 |
| Costa Rica          | 17.6              | 15.1                  | 7.3                 | 10.2  | 18.9  |
| Dominican Republic  | 7.8               | 8.6                   | 7.5                 | 3.8   | 12.0  |
| Ecuador             | 26.1              | 24.8                  | 9.7                 | 18.9  | 28.9  |
| Guatemala           | 20.9              | 18.6                  | 7.0                 | 13.8  | 22.4  |
| Haiti               | -5.0              | 0.6                   | 6.4                 | -4.4  | 4.8   |
| Honduras            | 6.6               | 6.8                   | 7.0                 | 2.0   | 10.5  |
| Jamaica             | 16.9              | 8.5                   | 6.0                 | 4.5   | 12.0  |
| Mexico              | -17.8             | -16.9                 | 5.0                 | -20.0 | -14.5 |
| Nicaragua           | 7.8               | 7.9                   | 6.8                 | 3.5   | 11.9  |
| Panama              | 12.0              | 11.8                  | 8.4                 | 6.5   | 15.6  |
| Paraguay            | 7.1               | -1.6                  | 7.5                 | -6.5  | 2.0   |
| Peru                | 0.7               | -2.2                  | 6.7                 | -6.1  | 0.8   |
| Salvador            | 8.1               | 9.7                   | 6.8                 | 4.8   | 13.6  |
| Suriname            | 30.9              | 26.6                  | 7.3                 | 22.0  | 31.3  |
| The Bahamas         | 15.1              | 12.8                  | 6.7                 | 8.6   | 16.6  |
| Trinidad and Tobago | 25.1              | 18.6                  | 7.8                 | 13.0  | 22.9  |
| Uruguay             | 4.0               | 0.8                   | 8.0                 | -4.2  | 4.3   |
| Average LAC         | 7.4               | 5.7                   | 6.9                 | 1.2   | 9.2   |
| Average RoW         | 0.9               | 0.2                   | 6.3                 | -3.6  | 2.9   |
| Average World       | 2.1               | 1.2                   | 6.4                 | -2.7  | 4.1   |

Table 3: Change in Exchange Rates at the Product Level: May 2014 - February 2016

Note: Column (1) reports weighted means of the change in the productlevel AREER, where the weights are the share of each product in country's exports. Columns (2)-(5) report unweighted statistics. The last three rows show the simple average of the figures at the country level.

### Online Appendix "Competition-Adjusted Measures of Real Exchanges Rates"

by Stein, Fernandez, Rosenow, and Zuluaga

### A Mathematical Appendix

### A.1 AREER at the Country-Product Level

In Subsection 2.2 we claimed that equations (14) and (15) are equivalent under the following two assumptions:

- 1. The ratio between the gross output and value added of product p ( $y_{pj}^*$  and  $y_{pj}$ , respectively) is equal to  $\xi_p$  for all origins, that is,  $\frac{y_{pj}^*}{y_{pj}} = \xi_p > 1$  for all j.
- 2. Product p is not used as an input in its own production.

The above results can be proved as follows:

$$\begin{aligned} \alpha_{pji} &= \frac{y_{pj} - \tau_{pj}\kappa_{pj}x_{pj}}{y_{pj} - \tau_{pj}\kappa_{pj}x_{pj} + m_{pj,-i}} \\ &= \frac{\xi_p y_{pj} - \xi_p \tau_{pj}\kappa_{pj}x_{pj}}{\xi_p y_{pj} - \xi_p \tau_{pj}\kappa_{pj}x_{pj} + \xi_p m_{pj,-i}} \\ &= \frac{y_{pj}^* - x_{pj}}{y_{pj}^* - x_{pj} + \xi_p m_{pj,-i}}. \end{aligned}$$

The third line follows from assumption 1, and the result that  $\xi_p \tau_{pj} \kappa_{pj} x_{pj}$  is equal to  $x_{pj}$ . To see this last equality, note that  $x_{pj}$  can be decomposed into four terms: (1) the value added in exports that is originated in j's industry p, that is,  $\tau_{pj} \kappa_{pj} x_{pj}$ ; (2) the value added in exports originating from the industry p of other countries,  $(1 - \tau_{pj}) \kappa_{p,-j} x_{pj}$ , (3) the value of local intermediate inputs,  $\tau_{pj} (1 - \kappa_{pj}) x_{pj}$ , and (4) the value of foreign intermediate inputs,  $(1 - \tau_{pj}) (1 - \kappa_{p,-j}) x_{pj}$ . From assumption 2, it follows that  $(1 - \tau_{pj}) \kappa_{p,-j} x_{pj}$ is equal to zero (since product p is not used as an input in its own production) and, therefore, all the value added in  $x_{pj}$  originated in industry p should be sourced in country j, which by the assumption 1 implies that  $\xi_p \tau_{pj} \kappa_{pj} x_{pj}$  is equal to  $x_{pj}$ .

### A.2 AREER at the Country-Destination Level

In subsection 2.3 we claimed that the country-level AREER can be calculated as the weighted (geometrical) average of the country-destination-level AREERs, with the weights assigned to each destination being the traditional real exchange rate weights. To show that, we start rewriting equation (16) for all the destination for which country i exports as follows:

$$\left(AREER_{i}^{j}\right)^{\frac{X_{ij}}{X_{i}}} = RER_{ij}^{\alpha_{ji}\frac{X_{ij}}{X_{i}}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ik}^{(1-\alpha_{ji})\frac{M_{jk}}{M_{j,-i}}\frac{X_{ij}}{X_{i}}}, \forall j \neq i.$$

$$(18)$$

Multiplying the n-1 equations defined in (18), we have the following expression:

$$\begin{split} \prod_{\substack{j=1\\j\neq i}}^{n} \left(AREER_{i}^{j}\right)^{\frac{X_{ij}}{X_{i}}} &= \prod_{\substack{j=1\\j\neq i}}^{n} \left(RER_{ij}^{\alpha_{ji}\frac{X_{ij}}{X_{i}}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ik}^{(1-\alpha_{ji})\frac{M_{jk}}{M_{j,-i}\frac{X_{ij}}{X_{i}}}}\right) \\ &= \prod_{\substack{j=1\\j\neq i}}^{n} \left(RER_{ij}^{\alpha_{ji}\frac{X_{ij}}{X_{i}}} \prod_{\substack{k=1\\k\neq i,j}}^{n} RER_{ij}^{(1-\alpha_{ki})\frac{M_{kj}}{M_{k,-i}\frac{X_{ik}}{X_{i}}}}\right) \\ &= \prod_{\substack{j=1\\j\neq i}}^{n} \left(RER_{ij}^{\alpha_{ji}\frac{X_{ij}}{X_{i}}} + \sum_{\substack{k=1\\k\neq i,j}}^{n} (1-\alpha_{ki})\frac{M_{kj}}{M_{k,-i}\frac{X_{ik}}{X_{i}}}\right) \\ &= \prod_{\substack{j=1\\j\neq i}}^{n} RER_{ij}^{w_{ij}^{1}}. \end{split}$$

The second line is obtained after expanding the product and rearranging the factors for the real exchange rate of each country j. The third line groups the components associated with each trade partner. Finally, the fourth line follows directly from equations (2) and (4) in the main text.

### **B** Imputation Method

Since our equation for the country-product-level AREER requires having information for all the variables included in the computation, it was necessary to impute  $\alpha_{pji}$  for all cases with missing data. We use the following four-step imputation procedure:

- 1. Compute country j's local absorption of p as follows:  $\alpha_{pj} = \frac{y_{pj}^* x_{pj}}{y_{pj}^* x_{pj} + \xi_p m_{pj}}$ . Notice that this expression allows reducing (temporarily) the dimensionality of the procedure (from three to two dimensions).
- 2. Estimate the residuals  $r_{pj} = \alpha_{pj} \gamma_p$ , where  $\gamma_p$  is the in-sample mean of  $\alpha_{pj}$  for product p (the estimated fixed effect associated with p).
- 3. Build a predictive model for  $r_{pj}$  using information on exports  $(\ln x_{pj} \text{ and } x_{jp}/X_j)$ , imports  $(\ln m_{pj} \text{ and } m_{jp}/M_j)$ , local value added in exports  $(\tau_{pj})$ , country-level absorption  $(\alpha_j)$ , year fixed effects and regional fixed effects.<sup>38</sup> We do not include product fixed effects in this step to not overfit the data.
- 4. Using the results from the previous two steps, compute equation (15).

To improve predictive accuracy, we use boosted regression as described in Hastie et al. (2009) and James et al. (2013) in the fourth step. Boosted regression is a machine learning algorithm that sequentially fits regression trees, where the information used to estimate each model comes from the updated residuals of previous iterations and from a random subsample of the training data (bagging). Predictions from this method are

<sup>&</sup>lt;sup>38</sup>We include dummies for the following regions: East Europe, Western Europe, Latin American and the Caribbean, Middle East and North Africa, South Saharan Africa, South Asia, and East Asia and the Pacific. The base category is North America.

then constructed as the (weighted) sum of the estimations from all individual trees. In comparison to a linear model, boosted regression is a more flexible algorithm, well suited to capture non-linear relationships between the outcome and the predictors without overfitting the data. Additionally, boosting has outperformed other methods in terms of predictive accuracy (see, for example, the simulation study reported in Schonlau (2005) or the results described in Hastie et al. (2009)).

### C AREER at the Country Level: Differences in Competitive Structure

Table A1 provides an overview by exporter of the differential structure of competitors between the AREER indicators and the REER.

[Table A1 about here.]

### D AREER at the Country Level: Implication for AREER

Table A2 provides an overview by exporter on how changes in these weights influence the AREER vis-àvis the REER between May 2014 and February 2016. Columns 3 and 4 report variations of the AREER, computing weights for manufactured goods (AREER-M) and adjusting them additionally for commodities (AREER-MC).

[Table A2 about here.]

|                     | (1)   | (2)     | (3)      |
|---------------------|-------|---------|----------|
|                     | AREER | AREER-M | AREER-MC |
| Argentina           | 21.8  | 34.1    | 29.1     |
| Barbados            | 34.8  | 37.9    | 45.6     |
| Bolivia             | 15.6  | 52.9    | 64.4     |
| Brazil              | 17.9  | 30.2    | 34.0     |
| Chile               | 19.5  | 33.0    | 46.8     |
| Colombia            | 36.3  | 40.9    | 63.1     |
| Costa Rica          | 22.9  | 24.6    | 26.5     |
| Dominican Republic  | 29.0  | 33.1    | 44.1     |
| Ecuador             | 47.7  | 49.6    | 67.8     |
| Guatemala           | 24.7  | 25.9    | 49.5     |
| Haiti               | 48.5  | 55.4    | 56.6     |
| Honduras            | 31.4  | 31.7    | 39.6     |
| Jamaica             | 33.0  | 39.7    | 67.4     |
| Mexico              | 28.0  | 26.9    | 35.9     |
| Nicaragua           | 31.8  | 37.7    | 49.7     |
| Panama              | 36.6  | 36.1    | 39.8     |
| Paraguay            | 35.7  | 39.1    | 51.8     |
| Peru                | 23.2  | 31.0    | 45.2     |
| Salvador            | 24.9  | 26.1    | 36.6     |
| Suriname            | 42.9  | 60.7    | 73.7     |
| The Bahamas         | 43.4  | 40.4    | 49.0     |
| Trinidad and Tobago | 43.6  | 35.4    | 62.3     |
| Uruguay             | 23.9  | 35.1    | 40.5     |
| Average LAC         | 31.2  | 37.3    | 48.6     |
| Average RoW         | 32.8  | 33.4    | 43.5     |
| Average World       | 32.5  | 34.1    | 44.5     |

Table A1: Change in the structure of weight: AREER vs. REER

Note: Column (1) shows the change in the structure of weights when comparing the AREER with the (traditional) REER. Column (2) shows the same difference when the AREER is computed with only manufacturing goods and column (3) when commodities are included following the method proposed by Bayoumi et al. (2006). The last three rows show the simple average of the figures at the country level.

|                     | (1)   | (2)     | (3)      |
|---------------------|-------|---------|----------|
|                     | AREER | AREER-M | AREER-MC |
| Argentina           | 0.9   | 4.9     | 2.5      |
| Barbados            | 5.9   | 4.7     | 7.9      |
| Bolivia             | 0.8   | -11.6   | -0.2     |
| Brazil              | 0.9   | 0.9     | 3.1      |
| Chile               | 2.1   | 3.3     | 7.1      |
| Colombia            | 3.1   | 2.6     | 5.4      |
| Costa Rica          | 0.4   | 0.1     | 1.3      |
| Dominican Republic  | 2.2   | 0.6     | 3.4      |
| Ecuador             | 12.7  | 11.5    | 4.3      |
| Guatemala           | 3.5   | 1.1     | 9.0      |
| Haiti               | 5.6   | 4.9     | 5.5      |
| Honduras            | 3.2   | 1.6     | 6.3      |
| Jamaica             | 1.2   | -5.0    | 7.7      |
| Mexico              | 2.8   | 2.2     | 4.2      |
| Nicaragua           | 2.3   | 1.9     | 6.1      |
| Panama              | 2.1   | 3.2     | 4.2      |
| Paraguay            | -0.8  | -8.2    | -7.6     |
| Peru                | 2.0   | 1.5     | 4.2      |
| Salvador            | 3.3   | 2.6     | 6.9      |
| Suriname            | -0.0  | 0.9     | 7.9      |
| The Bahamas         | 3.0   | 4.4     | 4.2      |
| Trinidad and Tobago | 5.9   | -2.4    | -2.1     |
| Uruguay             | 1.7   | 2.1     | 1.0      |
| Average LAC         | 2.8   | 1.2     | 4.0      |
| Average RoW         | -0.7  | -0.8    | 0.6      |
| Average World       | -0.1  | -0.4    | 1.3      |

Table A2: Change in Exchange Rates: May 2014 - February 2016

Note: Column (1) shows the change in competitiveness from May 2014 to February 2016 when comparing the AREER with the (traditional) REER. Column (2) shows the same difference when the AREER is computed with only manufacturing goods and column (3) when commodities are included following the method proposed by Bayoumi et al. (2009). The last three rows show the simple average of the figures at the country level.